Author’s response to reviews

Title: Knowledge of obstetric danger signs and associated factors: a study among mothers in Shashemene town, Oromia region, Ethiopia

Authors:

Biresh Wassihun (bireswas@gmail.com)
Berhanu Negese (berhanu.negese@gmail.com)
Hunduman Bedada (hundebedada@gmail.com)
Solomon Bekele (bekelesolomon788@gmail.com)
Agegnehu Bante (agegnehubante@gmail.com)
Tomas Yeheyis (tyeheyis@gmail.com)
Agere Abebe (age2098@gmail.com)
Duro uli (droulibuta@gmail.com)
Merima Mohammed (Merema.m2018@gmail.com)
Salasebish Gashawbez (Salasebesh@gmail.com)
Emebet Hussen (emebethussen@gmail.com)

Version: 2 Date: 23 Jan 2019

Author’s response to reviews:

all of the comments was Incorporated according to reviewers comments and suggestion